Author’s response to reviews

Title: Physical activity program for patients with dementia and their relative caregivers. Randomized clinical trial in Primary Health Care. AFISDEMyF.

Authors:

Emiliano Rodriguez-Sanchez MD (emilianorodriguezsanchez@yahoo.es)
José María Criado-Gutiérrez MD (jmcriado@usal.es)
Sara Mora-Simón PhDStudent (saramorasimon@gmail.com)
M Paz Muriel-Diaz MD (pazmd@ono.com)
Manuel A Gómez-Marcos MD (magomez@usal.es)
José I Recio-Rodríguez MD (donrecio@gmail.com)
M Carmen Patino-Alonso PhD (carpatino@usal.es)
Luis Valero-Juan MD (luva@usal.es)
José A Maderuelo-Fernandez MD (jmaderuelo@saludcastillayleon.es)
Luis García-Ortiz MD (Lgarciao@usal.es)

Version: 2
Date: 27 February 2014

Author’s response to reviews:

The modifications are described in correspondence with each of the editor’s comments:

1.- We have sent the email to BMCSeriesEditorial@biomedcentral.com. The documentation of the ethics committee approval and funding: the original and the translated documents with recognition from the institution (stamp and signature).

2.- We have modified the manuscript and added information on the status of the Study:
   Trial Status: The trial is currently in the recruitment phase.

3.- There are currently no other published articles on this study.